1	Unravelling genetic causality of haematopoiesis on bone metabolism in human
2	Shun-Cheong HO <sup>1#</sup> , Gloria Hoi-Yee LI <sup>2#</sup> , Anskar Yu-Hung LEUNG <sup>3</sup> , Kathryn Choon-Beng
3	TAN <sup>3</sup> , and Ching-Lung CHEUNG <sup>1,4</sup> *
4	<sup>#</sup> Co-first Authors
5	<sup>1</sup> Department of Pharmacology and Pharmacy, LKS Faculty of Medicine, The University of
6	Hong Kong, <sup>2</sup> Department of Health Technology and Informatics, Faculty of Health and
7	Social Sciences, The Hong Kong Polytechnic University, <sup>3</sup> Department of Medicine, Queen
8	Mary Hospital, LKS Faculty of Medicine, The University of Hong Kong, <sup>4</sup> Laboratory of Data
9	Discovery for Health (D <sup>2</sup> 4H), Hong Kong Science Park, Pak Shek Kok, Hong Kong
10	
11	Correspondence and reprint requests:
12	Ching-Lung Cheung, PhD
13	Department of Pharmacology and Pharmacy, LKS Faculty of Medicine, The University of
14	Hong Kong, 21 Sassoon Road, Pokfulam, Hong Kong
15	Email: lung1212@hku.hk
16	Tel: +852-3917-9462 Fax: +852-2816-2095
17	
18	Short title: Haematopoietic effect on bone metabolism
19	Keywords: Haematopoiesis, Bone Mineral Density, Fracture, Mendelian Randomization
20	Word Count: 245(Abstract); 3475 (Main text)
21	Number of Tables: 3, Number of Figures: 1

#### 22 Abstract

*Objective:* Haematopoiesis was shown to regulate bone metabolism in in vivo studies. However,
whether haematopoiesis has causal effects on bone health has never been investigated in human.
We aimed to evaluate the causal relationships of blood traits with bone mineral density (BMD)
and fracture.

*Design and Methods:* Using two-sample Mendelian randomization, causal relationship of 29
blood traits with estimated BMD (eBMD), total body BMD (TBBMD), lumbar spine BMD
(LSBMD), femoral neck BMD (FNBMD) and fracture were evaluated by inverse-variance
weighted (IVW) method and multiple sensitivity analyses. Relevant genetic data were obtained
from largest possible publicly available genome-wide association studies.

32 Results: Eight genetically determined red blood cell traits showed positive causal effects on eBMD, with beta estimates ranging from 0.009 (mean corpuscular haemoglobin) to 0.057 33 (haemoglobin concentration), while three white blood cell traits, including lymphocyte count 34 (beta:-0.020; 95% CI: -0.033 to -0.007), neutrophil count (beta:-0.020; 95% CI:-0.035 to -0.006) 35 and white blood cell count (beta:-0.027; 95% CI:-0.039 to -0.014), were inversely associated 36 37 with eBMD. Causal effects for six of these blood traits were validated on TBBMD, LSBMD, FNBMD and/or fracture. The association of reticulocyte count (beta:0.040; 95% CI:0.016 to 38 0.063), haemoglobin (beta: 0.058; 95% CI: 0.021 to 0.094) and mean corpuscular haemoglobin 39 concentration (beta:0.030; 95% CI:0.007 to 0.054) with eBMD remained significant in 40 multivariable IVW analyses adjusted for other blood traits. 41

42 *Conclusion:* This study provided evidence that haematopoietic system might regulate skeletal
43 system in human and suggested the possible pathophysiology of bone diseases among people
44 with haematological diseases.

## 46 Significance statement

We conducted a novel Mendelian randomization study investigating causal relationship of 47 blood cells with bone mineral density. Red and white blood cell traits have positive and inverse 48 49 causal relationship with bone mineral density respectively, suggesting a potential link of haematopoietic system with skeletal system in human. Current findings suggest individuals 50 with related haematological diseases, such as anaemia and leukocytosis, may have a lifelong 51 increased risk of osteoporosis and/or fracture. Given that complete blood count is commonly 52 performed in clinical setting, whether complete blood count can be used to predict fracture risk 53 54 warrants further investigation.

### 55 Introduction

In the last decade, bone cells were shown to be a novel regulator of haematopoiesis, such as 56 providing and interacting with haematopoietic stem cell niche (1-3), as well as secreting 57 erythropoietin (EPO) (4). Bone tissue also forms part of the osteoimmune system, at which 58 progenitors of myeloid and lymphoid cells, and mature immune cells are maintained (5). 59 Meanwhile, haematopoiesis itself generates various types of blood cells, including white blood 60 cells, which contribute to innate and acquired immune responses. As the key player of the 61 immune system, the count of total white blood cells, as well as the count of each type of white 62 blood cells, serve as the major diagnostic tools for disorders of the immune system, 63 inflammatory and infectious diseases (6). The blood cell traits therefore enabled, at least in part, 64 investigation of the human immune system. Although it is well accepted that there is interplay 65 66 among haematopoietic, immune and skeletal systems in the field of osteoimmunology, whether haematopoiesis and immune systems are causally linked to bone metabolism remains largely 67 68 unknown.

69

In terms of cell lineage, it is well known that osteoclasts are members of the 70 monocyte/macrophage lineage. While osteoblasts are generally derived from mesenchymal 71 stem cells, a recent in vivo study demonstrated that haematopoietic stem cells can give rise to 72 osteoblasts in a murine model (7). However, the relationship between blood traits and bone 73 metabolism in human is unclear. Although a few studies have evaluated the relationship of 74 blood traits with bone mineral density (BMD), conflicting results were observed for white 75 76 blood cells (8-10), red blood cells (8-13), and platelets (8, 9), which could be due to reverse causation and unmeasured confounding that are commonly encountered in conventional 77 observational studies. 78

80 Mendelian randomization (MR) has been gaining popularity in recent years as its study design

might overcome such limitations of epidemiological observational studies. In this two-sample MR study, we aimed to evaluate the causal relationship of 29 blood traits with BMD at heel estimated by ultrasound (eBMD), BMD at total body (TBBMD), lumbar spine (LSBMD) and femoral neck (FNBMD) measured by dual energy X-ray absorptiometry (DXA), as well as fracture, in the univariable analysis. Genetic correlation among the blood traits were calculated, and subsequent multivariable MR analyses were performed to evaluate the independent causal effects of blood traits on BMD.

88

## 89 Methods

#### 90 *Study design*

The study design and key assumptions of univariable MR analysis is shown in Fig. 1. Genetic variants associated with the exposure were utilized as instrumental variables (IVs) to infer causality with the outcome. Two-sample MR approach was adopted to evaluate causal relationship of various blood traits on BMD measured at different skeletal sites, including eBMD, TBBMD, LSBMD and FNBMD, as well as fracture. A detailed description of data sources for all the exposures and outcomes are presented in Table 1.

97

### 98 *Data source*

99 Exposure for MR analyses were 29 blood cell traits, with summary statistics retrieved from a
100 genome-wide association study (GWAS) comprising 408,112 UK Biobank participants of
101 European ancestry as the discovery cohort (14). Independent genome-wide significant genetic

variants associated with blood traits identified by the GWAS were selected as IVs (details in
Supplementary Methods A, <u>https://osf.io/k4m37/</u> (15)). The outcome was eBMD in primary
analyses, with summary statistics obtained from the largest GWAS of BMD to-date that
comprised 426,824 UK Biobank participants (16). Secondary MR analyses evaluating the
causal effects of blood traits on other bone traits (including TBBMD, LSBMD and FNBMD
and fracture) were conducted to validate the results from primary analysis (details described in
Supplementary Methods B, <u>https://osf.io/k4m37/</u> (15)).

109

## 110 Univariable MR analysis

The main analysis for the MR study was the inverse-variance weighted (IVW) method, which 111 112 assumed all IVs are valid (17). Prior to main analysis, radial MR analysis was used to identify outliers which had large contribution to Cochran's Q statistics that might imply heterogeneity, 113 including horizontal pleiotropy (18). MR Steiger filtering method was adopted to calculate and 114 compare the variance explained by all IVs in exposure and outcome, and infer the direction of 115 causality for each IV. IVs identified to have the expected causal direction from exposure (blood 116 117 traits) to outcome (bone traits) were kept (19). Using the same principle, MR Steiger directionality test was also applied to orient the overall causal effect of all IVs (19). Online 118 calculators were utilized for power calculation (20) and measuring strength of IVs. Summary 119 of the IVs adopted in various analyses are presented in Supplementary Tables 1-5 120 (https://osf.io/k4m37/ (15)). Sensitivity analyses, including weighted median (21), MR-Egger 121 (22) and contamination mixture (23) methods, were applied (details in Supplementary Methods 122 123 C, <u>https://osf.io/k4m37/</u> (15)). A causal relationship was only considered genuine if significant association was demonstrated in IVW [multiple testing corrected by false discovery date (FDR) 124 q-value <0.05], weighted median, and contamination mixture methods, while MR-Egger 125

intercept test was insignificant. To ensure the second assumption (independence assumption,
Fig. 1) holds, supplementary analysis was performed by excluding IVs associated with
potential confounding factors if significant causal relationship was identified for the primary
analysis of eBMD (details in Supplementary Methods D, <u>https://osf.io/k4m37/</u> (15)).

130

### 131 *Genetic correlation and multivariable MR*

In view of the presence of calculated and compound blood traits, genetic correlation between traits were calculated to derive the minimal representative set of blood traits (Supplementary Methods E, <u>https://osf.io/k4m37/</u> (15)). If any of the representative blood traits was shown to casually affect eBMD in univariable analysis, multivariable IVW analysis was performed by adjusting for the beta estimates of other representative blood traits (Supplementary Methods F, https://osf.io/k4m37/ (15)).

138

## 139 **Results**

## 140 *<u>Primary univariable analyses</u>*

The characteristics of the IVs are presented in Supplementary Table 1 (https://osf.io/k4m37/ 141 (15)). Out of 29 blood traits examined, 11 showed possible causal relationships with eBMD in 142 IVW analysis (FDR q-value<0.05), with similar results yielded in sensitivity analyses. Eight 143 of the 11 traits were red blood cell traits having positive causal effects on eBMD. Four were 144 immature red blood cell traits, including high light scatter reticulocyte count [IVW: per SD 145 increase in genetically determined blood trait, beta estimate in eBMD (in SD): 0.024; 95% CI: 146 0.013 to 0.035], high light scatter reticulocyte percentage (0.022; 95% CI: 0.011 to 0.034), 147 reticulocyte count (0.031; 95% CI: 0.019 to 0.042) and reticulocyte percentage 0.029; 95% CI: 148 0.018 to 0.040) (Table 2a and Supplementary Fig. 1, https://osf.io/k4m37/ (15)). The four 149

150 mature red blood cell traits were haematocrit (0.046; 95% CI: 0.031 to 0.062), haemoglobin concentration (0.057; 95% CI: 0.041 to 0.073), mean corpuscular haemoglobin (0.009; 95% 151 CI: 0.001 to 0.018), and mean corpuscular haemoglobin concentration (0.036; 95% CI: 0.020 152 to 0.051) (Table 2b and Supplementary Fig. 2, https://osf.io/k4m37/ (15)). Three white blood 153 cell related traits, including lymphocyte count (-0.020; 95% CI: -0.033 to -0.007), neutrophil 154 count (-0.020; 95% CI: -0.035 to -0.006) and white blood cell count (-0.027; 95% CI: -0.039 155 to -0.014), showed inverse causal effects on eBMD (Table 2c and Supplementary Fig. 3, 156 https://osf.io/k4m37/ (15)). Evidence of association was unavailable for other red blood cells, 157 158 white blood cells and platelet-related traits (Supplementary Table 6 and Supplementary Fig. 1-4, https://osf.io/k4m37/ (15)). The potential bias and type I error rate incurred by the sample 159 overlap was minimal as presented in Supplementary Table 1 (https://osf.io/k4m37/ (15)). 160

161

To ensure the independence assumption holds for the IVs, a supplementary analysis was 162 conducted by excluding IVs which are associated with potential confounders (listed in 163 Supplementary Table 7, https://osf.io/k4m37/ (15)). Among the 11 blood traits significantly 164 associated with eBMD in primary analysis, significant causal effects of high light scatter 165 166 reticulocyte count [IVW: per SD increase in genetically determined blood trait, beta estimate in eBMD (in SD): 0.068; 95% CI: 0.038 to 0.098], reticulocyte count (0.073; 95% CI: 0.042 to 167 0.104) and haematocrit (0.092; 95% CI: 0.064 to 0.12) with eBMD were consistently observed 168 in IVW and other sensitivity analysis, with larger effect sizes (Supplementary Table 8, 169 https://osf.io/k4m37/ (15)). For other blood cell traits, the causal associations with eBMD were 170 attenuated, or inconsistent associations between IVW and sensitivity analyses were found, 171 which could be attributed to the reduced statistical power upon exclusion of IVs 172 (Supplementary Table 9, https://osf.io/k4m37/ (15)). 173

174

## 175 *Genetic correlation and multivariable analyses*

Based on the pairwise genetic correlation of the blood traits (Supplementary Table 10, 176 https://osf.io/k4m37/ (15)), a minimal set of 21 traits were selected to represent all the 29 blood 177 traits (Supplementary Table 11, https://osf.io/k4m37/(15)). Among the 21 representative traits, 178 reticulocyte count, haemoglobin concentration, mean corpuscular haemoglobin, mean 179 corpuscular haemoglobin concentration, lymphocyte count and white blood cell count were 180 causally associated with eBMD in univariable analyses (Table 2). Thus, multivariable MR 181 analyses were performed for these traits to evaluate whether their causal effects on eBMD were 182 independent of other blood traits, by conditioning on the other 20 representative blood traits. 183

184

Multivariable IVW analyses demonstrated little change of causal estimates for reticulocyte count (0.040; 95% CI: 0.016 to 0.063), haemoglobin (0.058; 95% CI: 0.021 to 0.094) and mean corpuscular haemoglobin concentration (0.030; 95% CI 0.007 to 0.054) on eBMD. The multivariable MR-Egger test yielded similar positive causal estimates (Table 3). Nevertheless, the causal relationship of mean corpuscular haemoglobin, lymphocyte count and white blood cell count with eBMD were attenuated in the multivariable analysis. All the multivariable MR-Egger intercept tests were insignificant (Table 3).

192

## 193 <u>Secondary analyses</u>

Secondary analyses were performed to evaluate the causal relationship of blood traits with other bone-related traits, including TBBMD, LSBMD, FNBMD and fracture. Characteristics of the IVs adopted in the secondary analyses are presented in Supplementary Tables 2-5 197 (https://osf.io/k4m37/ (15)). Among the 11 blood traits which were shown to have causal effects on eBMD in the univariable analyses (Table 2), the association was validated for two 198 white and two red blood cell traits with DXA-derived BMD. Genetically increased white blood 199 200 cell count was inversely associated with TBBMD (Supplementary Table 12, https://osf.io/k4m37/ (15)) while neutrophil count was inversely associated with LSBMD 201 (Supplementary Table 13, https://osf.io/k4m37/(15)). Both traits were associated with reduced 202 FNBMD (Supplementary Table 14, https://osf.io/k4m37/ (15)). For red blood cell traits, 203 reticulocyte percentage and high light scatter reticulocyte percentage were positively 204 205 associated with FNBMD (Supplementary Table 14, https://osf.io/k4m37/ (15)). Similarly, three of the 11 causal relationships were validated for fracture, including genetically increased 206 haematocrit and haemoglobin with reduced risk of fracture; and increased neutrophil count 207 208 with increased fracture risk (Supplementary Table 15, https://osf.io/k4m37/ (15)). The 209 potential bias and type I error rate incurred by the sample overlap between the GWAS of blood traits and fracture was minimal as presented in Supplementary Table 5 (https://osf.io/k4m37/ 210 (15)). 211

212

#### 213 Discussion

Our study findings provide evidence on the potential link between the haematopoietic and skeletal systems in human and the pathophysiology of bone diseases among people with haematological diseases. In primary analyses, genetically increase in red and white blood cell traits were observed to have positive and inverse causal effects on eBMD respectively (Supplementary Fig. 5, <u>https://osf.io/k4m37/</u> (15)). Genetically increased haemoglobin also had positive causal effect on eBMD independent of other blood traits, while consistent association with reduced fracture risk was observed.

Several observational studies have been conducted to evaluate the relationship of blood cell 222 traits with BMD. Positive (8, 10-12, 24), null (13), and inverse (9, 11) association were 223 observed for different blood cell traits. The discrepancies could be explained by the cross-224 sectional nature of study design, as well as differences in study participants, such as the general 225 226 population (13), non-anaemic population (12), post-menopausal women (8-10, 24) and old men (11). The discrepancies could also be due to the definition of blood cell traits. For example, 227 three studies (8-10) examined "white blood cell counts", which consisted of a mixed population 228 of white blood cells. Conversely, the MrOS study (11) investigated counts of different white 229 blood cells individually and showed that association of high neutrophil, low lymphocyte, and 230 low monocyte were associated with rapid bone loss. In addition, analysis bias and selection 231 232 bias might explain the different results. Unmeasured confounders, such as comorbidities, might cause biases (9). Meanwhile, MR approach was reported to be less subjected to residual 233 confounding and reverse causation when compared to conventional observational studies. We 234 adopted the two-sample MR approach using GWAS data of 29 blood traits to investigate the 235 problem and demonstrated that red blood cell and white blood cell traits had positive and 236 237 inverse causal effects on eBMD respectively.

238

Due to the sample overlapping problem in the primary analyses, we performed secondary analyses in samples with minimal overlap. Some of the causal relationships observed for eBMD (including reticulocyte percentage, high light scatter reticulocyte percentage, neutrophil count and white blood cell count) could be validated in IVW analyses for DXA-derived BMD. For the non-reproducible traits, one plausible reason was the relatively small sample size of the GWAS of DXA-derived BMD, and hence low statistical power. Assume the causal estimates

for DXA-derived BMD were the same as that observed for eBMD in primary analyses, 245 minimum power of 26.6% and 12.7% was observed for mean corpuscular haemoglobin with 246 TBBMD and LSBMD/FNBMD respectively (Supplementary Table 16, https://osf.io/k4m37/ 247 (15)). Another reason could be the intrinsic difference between eBMD and DXA-derived BMD. 248 eBMD was positively and modestly correlated with DXA-derived TBBMD, LSBMD and 249 FNBMD (r~0.4-0.6) (16). While most of the susceptibility loci of DXA-derived BMD were 250 also identified in the GWAS of eBMD, a few eBMD-associated loci were reported to have 251 opposite direction of effects when compared to the DXA-derived BMD traits (25). In addition, 252 253 genetically increased haematocrit and haemoglobin were shown to reduce fracture risk, whereas neutrophil count elevated fracture risk. Effects of these blood traits on fracture were 254 255 directionally consistent with their effects on BMD, although not all causal relationships could 256 be validated. This might be explained by the relatively small sample size of the fracture GWAS 257 and hence limited power (Supplementary Table 5, https://osf.io/k4m37/ (15)). Another reason might be the multifactorial nature of fracture, such as non-bone related factors like muscle 258 strength and propensity of falls. Nevertheless, these do not affect the findings from our MR 259 analyses that lifelong increase in several red blood cell traits might be protective to bone, while 260 lifelong increase in certain white blood cell traits might be harmful to bone health. 261

262

The protective and independent effects of red blood cell traits on bone metabolism are supported by laboratory studies and clinical observations. Haematopoietic progenitors secrete osteolectin that is important for bone mass maintenance during adulthood (26, 27). Clinically, patients with sickle cell disease are more likely to have low bone mass and fragility fracture (28). In addition, a very recent study showed that erythroferrone, which is secreted by bone marrow erythroblasts, is a positive regulator of bone metabolism (29).

On the other hand, our univariable MR analyses demonstrated that genetically increased counts 270 of lymphocyte, neutrophil and white blood cell had adverse effects on BMD, and increased 271 neutrophil count might even elevate the fracture risk. Cautious interpretation may be required, 272 as the causal association of these white blood cell traits with eBMD was no longer significant 273 in the supplementary analysis excluding IVs associated with potential confounders. 274 Nevertheless, due to removal of a substantial portion (up to 68.2%) of IVs, the variance 275 explained by the IVs on the exposure decreased, reducing the statistical power of the analysis. 276 We only have 37.2% to 65.3% power in detecting a 0.02 SD change in eBMD per SD increase 277 in white blood cell traits (Supplementary Table 9, https://osf.io/k4m37/ (15)). The analysis 278 should be re-visited when the power issue is resolved, such as having GWAS of larger sample 279 280 size. Lymphocyte and neutrophil account for over 90% of the five main types of white blood cells in human. Inflammation, which is characterized by increased white blood cell counts, is 281 a well-established risk factor of osteoporosis, due to its stimulation of osteoclastogenesis. 282 Autoantibodies and cytokine would stimulate the osteoclast differentiation. For example, 283 cytokines released from T-lymphocyte such as granulocyte macrophage-colony stimulating 284 285 factor (GM-CSF) could facilitate mature osteoclast formation (30, 31). CCL2 (C-C motif 286 chemokine ligand 2) and CXCL1 (C-X-C motif chemokine ligand 1), two chemokines secreted 287 from neutrophil, are mediators of osteoclastogenesis that can accelerate osteoclast maturation 288 (32). Nevertheless, lymphocyte count and white blood cell count were selected in the minimal set of 21 representative blood traits. Multivariable analyses by adjustment for other 289 representative blood traits attenuated the association with eBMD, implying that total effects of 290 291 these two traits on eBMD as observed in univariable analyses could be, in part, explained by its correlation with other blood traits. 292

294 There are clinical implications in the current study. Our findings suggested that people with related haematological diseases, such as anaemia and leukocytosis, may have a lifelong 295 increased risk of osteoporosis and/or fracture. Yet, the observed effect estimates of blood traits 296 297 on BMD were small, which could be due to the following reasons. First, eBMD was used in the primary analysis, which may not accurately reflect the effect size for DXA-measured BMD. 298 Although DXA-measured BMD was examined in the secondary analysis, the GWAS were of 299 small sample size and had insufficient power, potentially leading to inaccurate estimation of 300 effect size. Second, the GWAS of blood traits adopted in this MR study were measured in 301 302 healthy people without serious haematological disorders. The current MR findings could only reflect the effect of blood traits on BMD in people without haematological disorders, while it 303 304 is expected that the effect of haematological disorders on BMD could be even stronger. Third, 305 the estimates used in the analyses were adjusted for several covariates in the original GWAS, 306 thus the effect estimates observed in the study may reflect the sole effect of blood traits on BMD, which is small but statistically significant. It should be noted that, interventions that 307 affect the blood traits, especially the pharmacological ones, may also affect the skeletal systems. 308 Future studies examining the relationship of interventions affecting the blood traits with BMD 309 are warranted. On the other hand, given that complete blood count is commonly performed in 310 clinical setting, whether complete blood count can be used to predict long-term bone health in 311 terms of BMD and fracture risk warrants further investigation. 312

313

The strength of this study was the inclusion of summary statistics from the largest GWAS metaanalysis and large number of IVs in primary analyses that provided ample power. The primary analyses had at least 80% power in detecting a causal estimate of as low as 0.008 SD change in eBMD per SD change in several blood traits (Supplementary Table 1, <u>https://osf.io/k4m37/</u> (15)). Thus, even if a causal effect is present for these traits on eBMD, the genuine effect size may be too small to be detected in the present study. The study design of MR allowed causal inference between traits, which was often infeasible to be evaluated using randomized controlled trial. Adoption of a stringent definition of potentially genuine finding in this study reduces false positive rate.

323

Nevertheless, there were limitations. For the causal association revealed to be potentially 324 genuine in the primary analyses, MR-Egger intercept tests were all insignificant, implying 325 horizontal pleiotropy was unlikely, although it cannot be completely ruled out. Yet, a consistent 326 conclusion could be drawn from the supplementary analysis excluding IVs associated with 327 potential confounder and multivariable MR analysis that several red blood cell traits were 328 329 independently associated with BMD. Furthermore, both GWAS of blood cell traits and eBMD 330 comprised study participants from the UK Biobank, possibly leading to biases towards the direction of the observational association due to the weak instrument bias (33). In particular, 331 the IVs were discovered in the data adopted in MR analysis, which was reported to worsen the 332 issue of weak instrument bias due to winner's curse (33). Nevertheless, due to the relatively 333 high F-statistics of the IVs (≥97.06, average per IV; Supplementary Table 1, 334 https://osf.io/k4m37/ (15)), weak instrument bias was not expected despite sample overlap. A 335 recent simulation study also suggested that two-sample MR methods could be safely applied 336 within biobanks of large sample size (>300,000) (34). With the assumption that sample overlap 337 between the exposure and outcome datasets was 100% and the observational estimate was 0.1 338 SD change in eBMD per SD increase in the blood cell traits, the bias and Type 1 error rate due 339 to sample overlap under the null hypothesis were 0.001 SD in eBMD and 0.05 respectively 340 (Supplementary Table 1, <u>https://osf.io/k4m37/</u> (15)). Moreover, we conducted secondary 341 analyses to assess the causal effects of blood traits on DXA-derived BMD and fracture, which 342 had no or minimal sample overlap with the UK Biobank cohort. The secondary analyses 343

validated the causal effects of six blood traits on bone health. Therefore, the potentially genuine
causal relationships observed in this study were likely to be true. Lastly, GWAS/GWAS metaanalysis of DXA-derived BMD and fracture had relatively small sample size. The lack of
evidence of association might be due to insufficient statistical power.

348

In conclusion, this study suggested that both red blood cell and white blood cell parameters might causally influence BMD and fracture risk. While the current study evaluated the causal relationship of haematopoiesis with bone health using genetic data from the general population, future investigation on the pathophysiology between haematological diseases and bone health is warranted.

355 Declaration of interest: There is no conflict of interest that could be perceived as prejudicing356 the impartiality of the research reported.

Funding: This research did not receive any specific grant from any funding agency in the public,commercial or not-for-profit sector.

Author contributions: S.C.H., G.H.L., C.L.C. designed the study. S.C.H., G.H.L., C.L.C. collected data and conducted data analysis. S.C.H., G.H.L., C.L.C. interpreted the data and drafted the manuscript. S.C.H., G.H.L., A.Y.L., K.C.T., C.L.C. revised the manuscript. All authors approved final version of manuscript. C.L.C. takes responsibility for the integrity of the data analysis.

364 Acknowledgements: None

365 Data availability: The genetic data used in the current study were publicly available. Related366 sources were described in the manuscript.

367 Ethical approval: Ethical approval was available from original genome-wide association368 studies from which the genetic data was extracted for the current analysis.

369 Informed consent: Informed consent from participants was available from original genome-

370 wide association studies from which the genetic data was extracted for the current analysis.

## 371 **References**

- Omatsu Y, Sugiyama T, Kohara H, Kondoh G, Fujii N, Kohno K & Nagasawa T. The 1. 372 essential functions of adipo-osteogenic progenitors as the hematopoietic stem and progenitor 373 cell niche. Immunity 2010 33 387-399. (https://doi.org/10.1016/j.immuni.2010.08.017) 374 Ding L & Morrison SJ. Haematopoietic stem cells and early lymphoid progenitors 375 2. occupy distinct bone marrow niches. Nature 2013 495 231-235. 376 (https://doi.org/10.1038/nature11885) 377 Greenbaum A, Hsu YM, Day RB, Schuettpelz LG, Christopher MJ, Borgerding JN, 378 3. Nagasawa T & Link DC. CXCL12 in early mesenchymal progenitors is required for 379 380 haematopoietic stem-cell maintenance. Nature 2013 495 227-230. (https://doi.org/10.1038/nature11926) 381 Rankin EB, Wu C, Khatri R, Wilson TL, Andersen R, Araldi E, Rankin AL, Yuan J, 4. 382 Kuo CJ, Schipani E, et al. The HIF signaling pathway in osteoblasts directly modulates 383 erythropoiesis through the production of EPO. Cell 2012 149 63-74. 384 (https://doi.org/10.1016/j.cell.2012.01.051) 385 5. Tsukasaki M & Takayanagi H. Osteoimmunology: evolving concepts in bone-386 immune interactions in health and disease. Nat Rev Immunol 2019 19 626-642. 387 (https://doi.org/10.1038/s41577-019-0178-8) 388 389 Blumenreich MS. The White Blood Cell and Differential Count. In Clinical Methods: 6. The History, Physical, and Laboratory Examinations, edn 3, ch. 153, pp 724-727. Eds 390 391 Walker HK, Hall WD & Hurst JW. Boston: Butterworths, 1990. 392 7. Kang IH, Baliga UK, Wu Y, Mehrotra S, Yao H, LaRue AC & Mehrotra M. Hematopoietic stem cell-derived functional osteoblasts exhibit therapeutic efficacy in a 393 murine model of Osteogenesis imperfecta. Stem Cells 2021 39 1457-1477. 394 (https://doi.org/10.1002/stem.3432) 395 Kim HL, Cho HY, Park IY, Choi JM, Kim M, Jang HJ & Hwang SM. The positive 8. 396 association between peripheral blood cell counts and bone mineral density in postmenopausal 397 women. Yonsei Med J 2011 52 739-745. (https://doi.org/10.3349/ymj.2011.52.5.739) 398 Li L, Ge JR, Chen J, Ye YJ, Xu PC & Li JY. Association of bone mineral density 399 9. with peripheral blood cell counts and hemoglobin in Chinese postmenopausal women: A 400 retrospective study. Medicine 2020 99 e20906. 401 402 (https://doi.org/10.1097/md.000000000020906) Polineni S, Resulaj M, Faje AT, Meenaghan E, Bredella MA, Bouxsein M, Lee H, 10. 403 MacDougald OA, Klibanski A & Fazeli PK. Red and White Blood Cell Counts Are 404 405 Associated With Bone Marrow Adipose Tissue, Bone Mineral Density, and Bone Microarchitecture in Premenopausal Women. J Bone Miner Res 2020 35 1031-1039. 406 (https://doi.org/10.1002/jbmr.3986) 407 Valderrabano RJ, Lui LY, Lee J, Cummings SR, Orwoll ES, Hoffman AR, Wu JY & 408 11. Osteoporotic Fractures in Men Study Research G. Bone Density Loss Is Associated With 409 Blood Cell Counts. J Bone Miner Res 2017 32 212-220. (https://doi.org/10.1002/jbmr.3000) 410 12. Kim HS, Park HM, Lee HS & Lee YJ. Hemoglobin levels and low bone mineral 411 density in non-anemic older adults: Secondary analysis of the Korean National Health and 412
- 413 Nutrition Examination Survey. *Exp Gerontol* 2019 **126** 110706.
- 414 (<u>https://doi.org/10.1016/j.exger.2019.110706</u>)
- 415 13. Valderrabano RJ, Buzkova P, Chang PY, Zakai NA, Fink HA, Robbins JA, Lee JS,
- 416 Wu JY & Cardiovascular Health Study g. Association of bone mineral density with
- 417 hemoglobin and change in hemoglobin among older men and women: The Cardiovascular
- 418 Health Study. *Bone* 2019 **120** 321-326. (<u>https://doi.org/10.1016/j.bone.2018.11.010</u>)

419 14. Vuckovic D, Bao EL, Akbari P, Lareau CA, Mousas A, Jiang T, Chen MH, Raffield LM, Tardaguila M, Huffman JE, et al. The Polygenic and Monogenic Basis of Blood Traits 420 and Diseases. Cell 2020 182 1214-1231 e1211. (https://doi.org/10.1016/j.cell.2020.08.008) 421 Ho S, Li G, Leung A, Tan K & Cheung C. Unravelling genetic causality of 15. 422 haematopoiesis on bone metabolism in human - Supplementary Data. Available from: 423 https://osf.io/k4m37/ [Accessed 26 August 2022]. 424 425 16. Morris JA, Kemp JP, Youlten SE, Laurent L, Logan JG, Chai RC, Vulpescu NA, Forgetta V, Kleinman A, Mohanty ST, et al. An atlas of genetic influences on osteoporosis in 426 humans and mice. Nat Genet 2019 51 258-266. (https://doi.org/10.1038/s41588-018-0302-x) 427 428 17. Burgess S, Butterworth A & Thompson SG. Mendelian randomization analysis with multiple genetic variants using summarized data. Genetic epidemiology 2013 37 658-665. 429 430 (https://doi.org/10.1002/gepi.21758) Bowden J, Spiller W, Del Greco MF, Sheehan N, Thompson J, Minelli C & Davey 431 18. Smith G. Improving the visualization, interpretation and analysis of two-sample summary 432 data Mendelian randomization via the Radial plot and Radial regression. Int J Epidemiol 433 2018 47 1264-1278. (https://doi.org/10.1093/ije/dyy101) 434 Hemani G, Tilling K & Davey Smith G. Orienting the causal relationship between 435 19. imprecisely measured traits using GWAS summary data. PLoS genetics 2017 13 e1007081. 436 (https://doi.org/10.1371/journal.pgen.1007081) 437 438 20. Burgess S. Sample size and power calculations in Mendelian randomization with a 439 single instrumental variable and a binary outcome. Int J Epidemiol 2014 43 922-929. (https://doi.org/10.1093/ije/dyu005) 440 Bowden J, Davey Smith G, Haycock PC & Burgess S. Consistent estimation in 441 21. Mendelian randomization with some invalid instruments using a weighted median estimator. 442 Genetic epidemiology 2016 40 304-314. (https://doi.org/10.1002/gepi.21965) 443 444 22. Bowden J, Davey Smith G & Burgess S. Mendelian randomization with invalid instruments: effect estimation and bias detection through Egger regression. International 445 journal of epidemiology 2015 44 512-525. (https://doi.org/10.1093/ije/dyv080) 446 Burgess S, Foley CN, Allara E, Staley JR & Howson JMM. A robust and efficient 447 23. method for Mendelian randomization with hundreds of genetic variants. Nat Commun 2020 448 11 376. (https://doi.org/10.1038/s41467-019-14156-4) 449 Wu F, Ames R, Clearwater J, Evans MC, Gamble G & Reid IR. Prospective 10-year 450 24. study of the determinants of bone density and bone loss in normal postmenopausal women, 451 including the effect of hormone replacement therapy. Clin Endocrinol (Oxf) 2002 56 703-452 711. (https://doi.org/10.1046/j.1365-2265.2002.01534.x) 453 Kemp JP, Morris JA, Medina-Gomez C, Forgetta V, Warrington NM, Youlten SE, 454 25. Zheng J, Gregson CL, Grundberg E, Trajanoska K, et al. Identification of 153 new loci 455 associated with heel bone mineral density and functional involvement of GPC6 in 456 457 osteoporosis. Nat Genet 2017 49 1468-1475. (https://doi.org/10.1038/ng.3949) Zhang J, Cohen A, Shen B, Du L, Tasdogan A, Zhao Z, Shane EJ & Morrison SJ. The 458 26. effect of parathyroid hormone on osteogenesis is mediated partly by osteolectin. Proc Natl 459 Acad Sci U S A 2021 118 e2026176118. (https://doi.org/10.1073/pnas.2026176118) 460 Yue R, Shen B & Morrison SJ. Clec11a/osteolectin is an osteogenic growth factor 27. 461 that promotes the maintenance of the adult skeleton. *Elife* 2016 5 e18782. 462 (https://doi.org/10.7554/elife.18782) 463 Giordano P, Urbano F, Lassandro G & Faienza MF. Mechanisms of Bone Impairment 28. 464 in Sickle Bone Disease. Int J Environ Res Public Health 2021 18 1832. 465 466 (https://doi.org/10.3390/ijerph18041832) Castro-Mollo M, Gera S, Ruiz-Martinez M, Feola M, Gumerova A, Planoutene M, 29. 467 Clementelli C, Sangkhae V, Casu C, Kim SM, et al. The hepcidin regulator erythroferrone is 468

- a new member of the erythropoiesis-iron-bone circuitry. *Elife* 2021 **10** e68217.
- 470 (<u>https://doi.org/10.7554/elife.68217</u>)
- 471 30. Udagawa N. The mechanism of osteoclast differentiation from macrophages: possible
  472 roles of T lymphocytes in osteoclastogenesis. *Journal of bone and mineral metabolism* 2003
- 473 **21** 337-343. (https://doi.org/10.1007/s00774-003-0439-1)
- 474 31. Lee MS, Kim HS, Yeon JT, Choi SW, Chun CH, Kwak HB & Oh J. GM-CSF
- 475 regulates fusion of mononuclear osteoclasts into bone-resorbing osteoclasts by activating the
- 476 Ras/ERK pathway. *The Journal of Immunology* 2009 **183** 3390-3399.
- 477 (https://doi.org/10.4049/jimmunol.0804314)
- 478 32. Hu Y, Wang L, Zhao Z, Lu W, Fan J, Gao B, Luo Z, Jie Q, Shi X & Yang L.
- 479 Cytokines CCL2 and CXCL1 may be potential novel predictors of early bone loss. *Molecular* 480 *medicine reports* 2020 22 4716-4724. (<u>https://doi.org/10.3892/mmr.2020.11543</u>)
- 481 33. Burgess S, Davies NM & Thompson SG. Bias due to participant overlap in two-
- sample Mendelian randomization. *Genetic epidemiology* 2016 **40** 597-608.
- 483 (<u>https://doi.org/10.1002/gepi.21998</u>)
- 484 34. Minelli C, Del Greco M F, van der Plaat DA, Bowden J, Sheehan NA & Thompson J.
- The use of two-sample methods for Mendelian randomization analyses on single large datasets. *International journal of epidemiology* 2021 **50** 1651-1659.
- 487 (https://doi.org/10.1093/ije/dyab084)
- 488 35. Medina-Gomez C, Kemp JP, Trajanoska K, Luan J, Chesi A, Ahluwalia TS, Mook-
- 489 Kanamori DO, Ham A, Hartwig FP, Evans DS, et al. Life-Course Genome-wide Association
- 490 Study Meta-analysis of Total Body BMD and Assessment of Age-Specific Effects. *Am J* 491 *Hum Genet* 2018 102 88-102. (https://doi.org/10.1016/j.ajhg.2017.12.005)
- 492 36. Estrada K, Styrkarsdottir U, Evangelou E, Hsu YH, Duncan EL, Ntzani EE, Oei L,
- Albagha OM, Amin N, Kemp JP, *et al.* Genome-wide meta-analysis identifies 56 bone
- 494 mineral density loci and reveals 14 loci associated with risk of fracture. *Nat Genet* 2012 44 495 491-501. (https://doi.org/10.1038/ng.2249)
- 496 37. Trajanoska K, Morris JA, Oei L, Zheng HF, Evans DM, Kiel DP, Ohlsson C,
- 497 Richards JB, Rivadeneira F, consortium GG, et al. Assessment of the genetic and clinical
- determinants of fracture risk: genome wide association and mendelian randomisation study.
- 499 *BMJ* 2018 **362** k3225. (<u>https://doi.org/10.1136/bmj.k3225</u>)

500

Traits	Description	Ancestry	Sample size*	Derivation of independent genome-wide
				significant variants
Blood Traits (14)	A GWAS of 29 blood cell phenotypes, with a discovery cohort conducted in 408,112 UK BioBank participants. The 29 traits included 6 immature red blood cell traits, 8 mature red blood cell traits, 4 platelet-related traits and 11 white blood cell traits. A meta-analysis was also conducted with 154,976 additional European participants from the Blood Cell Consortium for 15 of the blood cell traits only. The full set of summary statistics for the discovery cohort, but not the meta-analysis, was publicly available online.	European	408,112 (Discovery cohort)	A total of 16,900 conditionally independent blood trait-variant associations were identified from the discovery cohort. Briefly, for each of the 29 blood traits, the genome-wide significant variants were partitioned into the largest number of blocks, with pairwise block separation of $\geq$ 5Mb. In each block, only a parsimonious set of variants were kept to explain the signals in the block using a stepwise multiple linear regression approach. In constructing the model, a new variant in high LD (r <sup>2</sup> >0.9) with a variant already in the parsimonious set would be ignored, as the new variant cannot represent an independent signal.
eBMD (16) (Primary analyses)	A GWAS of heel bone mineral density estimated by ultrasound (eBMD) in the white British UK Biobank cohort.	European	426,824	Conditional analysis was performed using GCTA, which used a standard stepwise selection model based on the LD pattern in the reference population. In brief, the analysis started with the variant with the strongest association in the meta- analysis. Other variants with minimum conditional p-value that fit all the selected variant(s) were subsequently added. Variants with large conditional p-values were excluded. The iteration stopped until no more variants were added or excluded. In this study, variants with $r^2>0.9$ were ignored, and the remaining variants situated >20Mbp away were defined as independent. 1,103 conditionally independent variants with genome-wide significance (p < 6.6x10 <sup>-9</sup> ) were identified.
TBBMD (35)	A GWAS meta-analysis of DXA-derived	86% European	66,628	Similar to the GWAS of eBMD, conditional
(Secondary analyses)	TBBMD from 30 cohorts across America, Europe, and Australia. The full set of summary statistics were available online.			analysis was performed using GCTA in the meta- analysis of cohorts with European ancestry only (N=56,284).

 Table 1 Data source used in Mendelian randomization analysis

Traits	Description	Ancestry	Sample size*	Derivation of independent genome-wide significant variants
				81 independent genome-wide significant (p $< 5x10^{-8}$ ) variants associated with TBBMD were identified in the meta-analysis with European-specific cohort.
LSBMD and FNBMD (36) (Secondary analyses)	A GWAS meta-analysis of DXA-derived LSBMD and FNBMD, with the discovery stage performed in 17 GWAS of populations across North America, Europe, East Asia and Australia, which were part of GEnetic Factors for OSteoporosis consortium (GEFOS). In the follow-up replication stage, de-novo genotyping of 96 independent genome-wide significant SNPs obtained from the discovery cohorts were performed in 50,933 additional participants from 34 cohorts. A meta-analysis of these 96 SNPs comprising the discovery and replication cohorts were also conducted. The full set of summary statistics were only publicly available for the discovery cohort.	Predominantly European (~ 70%)	32,961 (Discovery cohort)	At the discovery stage, 96 SNPs with the strongest association reaching genome-wide significance (p $< 5x10^{-8}$ ) were selected for replication. Out of the 96 replicated SNPs, 64 remained genome-wide significant in the meta-analysis of discovery and replication cohorts. The SNPs were regarded as independent if they were separated by $\ge 1$ Mb from the top signal. There were 48 and 49 independent genome-wide significant SNPs associated with LSBMD and FNBMD respectively.
Fracture (37) (Secondary analyses)	A GWAS meta-analysis of fracture. The discovery stage comprised 23 cohorts recruited globally through the GEFOS, which were predominantly of European descent and from Europe, North America, Australia, and east Asia. Two additional GWAS from UK Biobank and EPIC Norfolk study were included as an extended discovery dataset. Selected variants were replicated in 147,200 cases and 150,085 controls from 23andMe. The full set of summary statistics were only publicly available online for the discovery data, but not the meta- analysis of the discovery and replication data.	Predominantly European (~80%)	264,973 (case: 37,857; control: 227,116; Discovery cohort)	A total of 15 variants from 11 chromosomes achieved genome-wide significance ( $p < 5x10^{-8}$ ) in the meta-analysis of discovery and replication cohorts. For the variants on the same chromosome, they were separated by $\geq 9$ Mb, and were considered independent.

\* The sample size refers to the cohort from which the full set of summary statistics were obtained from.

**Table 2** Mendelian randomization results showing significant causal relationship of (a) immature red blood cell traits; (b) mature red blood cell traits; and (c) white blood cell traits with eBMD

## (a)

					MR-Egger int	ercept test	Mea overla	surement for apping sample
Methods	Estimate	95% CI	p-value	q-value	Estimate	p-value	Bias	Type I Error
High Light Scatter Reticul	ocyte Count							
IVW	0.024	0.013 - 0.035	2.16x10 <sup>-05</sup>	6.03x10 <sup>-05</sup>				
Weighted Median	0.019	0.006 - 0.031	0.004				0.001	0.0 <b>5</b>
MR-Egger	0.014	-0.004 - 0.033	0.127		4.26x10 <sup>-04</sup>	0.191	0.001	0.05
Contamination Mixture	0.04	0.03 - 0.05	3.72x10 <sup>-07</sup>					
High Light Scatter Reticul	ocyte Percentage							
IVW	0.022	0.011 - 0.034	9.67x10 <sup>-05</sup>	2.03x10 <sup>-04</sup>				
Weighted Median	0.020	0.007 - 0.033	0.003				0.001	0.05
MR-Egger	0.016	-0.003 - 0.034	0.097		2.99x10 <sup>-04</sup>	0.366	0.001	0.03
Contamination Mixture	0.04	0.03 - 0.05	2.15x10 <sup>-07</sup>					
Reticulocyte Count								
IVW	0.031	0.019 - 0.042	8.48x10 <sup>-08</sup>	4.74x10 <sup>-07</sup>				
Weighted Median	0.027	0.015 - 0.040	2.01x10 <sup>-05</sup>				0.001	0.05
MR-Egger	0.023	0.005 - 0.041	0.013		3.25x10 <sup>-04</sup>	0.315	0.001	0.03
Contamination Mixture	0.04	0.03 - 0.05	5.29x10 <sup>-10</sup>					
Reticulocyte Percentage								
IVW	0.029	0.018 - 0.040	2.37x10 <sup>-07</sup>	9.93x10 <sup>-07</sup>				
Weighted Median	0.023	0.011 - 0.035	2.78x10 <sup>-04</sup>				0.001	0.05
MR-Egger	0.020	0.002 - 0.038	0.027		4.01x10 <sup>-04</sup>	0.221	0.001	0.03
Contamination Mixture	0.04	0.03 - 0.04	6.35x10 <sup>-10</sup>					

					MR-Egger in	tercept test	Meas	surement for
Methods	Estimate	95% CI	p-value	q-value	Estimate	p-value	Bias	Type I Error
Haematocrit								
IVW	0.046	0.031 - 0.062	6.48x10 <sup>-09</sup>	5.43x10 <sup>-08</sup>				
Weighted median	0.038	0.020 - 0.055	4.06x10 <sup>-05</sup>				0.001	0.05
MR-Egger	0.022	-0.008 - 0.052	0.153		7.41x10 <sup>-04</sup>	0.061	0.001	0.03
Contamination Mixture	0.04	0.02 - 0.05	0.008					
Haemoglobin								
IVW	0.057	0.041 - 0.073	7.81x10 <sup>-13</sup>	1.31x10 <sup>-11</sup>				
Weighted median	0.055	0.037 - 0.073	1.66x10 <sup>-09</sup>				0.001	0.05
MR-Egger	0.038	0.009 - 0.067	0.011		6.07x10 <sup>-04</sup>	0.121	0.001	0.03
Contamination Mixture	0.04	0.03 - 0.05	5.91x10 <sup>-04</sup>					
Mean Corpuscular Haemog	globin							
IVW	0.009	0.001 - 0.018	0.032	0.042				
Weighted median	0.017	0.007 - 0.028	0.001				0.001	0.05
MR-Egger	0.018	0.005 - 0.031	0.006		-4.95x10 <sup>-04</sup>	0.071	0.001	0.03
Contamination Mixture	0.02	0.02 - 0.02	2.79x10 <sup>-06</sup>					
Mean Corpuscular Haemog	globin Concent	ration						
IVW	0.036	0.020 - 0.051	5.03x10 <sup>-06</sup>	1.68x10 <sup>-05</sup>				
Weighted median	0.041	0.023 - 0.059	8.38x10 <sup>-06</sup>				0.001	0.05
MR-Egger	0.053	0.027 - 0.080	8.62x10 <sup>-05</sup>		-7.44x10 <sup>-04</sup>	0.114	0.001	0.03
Contamination Mixture	0.05	0.04 - 0.06	4.37x10 <sup>-09</sup>					

					MR-Egger inte	ercept test	Meas	surement for
Methods	Estimate	95% CI	p-value	q-value	Estimate	p-value	Bias	Type I Error
Lymphocyte Count								
IVW	-0.020	-0.0330.007	0.002	0.004				
Weighted median	-0.025	-0.0410.010	0.001				0.001	0.05
MR-Egger	-0.021	-0.047 - 0.004	0.098		3.26x10 <sup>-05</sup>	0.929	0.001	0.03
Contamination Mixture	-0.04	-0.050.03	2.60x10 <sup>-04</sup>					
Neutrophil Count								
IVW	-0.020	-0.0350.006	0.006	0.010				
Weighted median	-0.033	-0.0500.015	2.74x10 <sup>-04</sup>				0.001	0.05
MR-Egger	-0.016	-0.043 - 0.011	0.253		-1.46x10 <sup>-04</sup>	0.703	0.001	0.03
Contamination Mixture	-0.03	-0.040.02	4.80x10 <sup>-05</sup>					
White Blood Cell Count								
IVW	-0.027	-0.0390.014	4. 50x10 <sup>-05</sup>	1.08x10 <sup>-04</sup>				
Weighted median	-0.039	-0.0540.024	4.88x10 <sup>-07</sup>				0.001	0.05
MR-Egger	-0.033	-0.0570.009	0.007		1.97x10 <sup>-04</sup>	0.555	0.001	0.03
Contamination Mixture	-0.05	-0.060.04	6.47x10 <sup>-10</sup>					

 Table 3 Multivariate Mendelian randomization results of blood traits with eBMD

					MR-Egger intercept test	
Exposure	Methods	Estimate	95% CI	p-value	Estimate	p-value
Immature Red Blood Cell Trait						
Reticulocyte Count						
	IVW	0.040	0.016 - 0.063	0.001		
	MR-Egger	0.035	0.008 - 0.062	0.010	0.000	0.477
<u>Mature Red Blood Cell Traits</u>						
Haemoglobin						
	IVW	0.058	0.021 - 0.094	0.002		
	MR-Egger	0.037	-0.006 - 0.081	0.093	0.001	0.090
Mean Corpuscular Haemoglobin						
	IVW	0.004	-0.007 - 0.016	0.450		
	MR-Egger	0.012	-0.003 - 0.027	0.112	0.000	0.115
Mean Corpuscular Haemoglobin Co	ncentration					
I I I I I I I I I I I I I I I I I I I	IVW	0.030	0.007 - 0.054	0.011		
	MR-Egger	0.049	0.014 - 0.085	0.006	-0.001	0.163
White Blood Cell Traits	66					
Lymphocyte Count						
<i>y i i j i i i i i i i i i i</i>	IVW	-0.021	-0.045 - 0.003	0.087		
	MR-Egger	-0.018	-0.049 - 0.013	0.263	0.000	0.775
White Blood Cell Count	00 -	•				
	IVW	-0.017	-0.043 - 0.009	0.197		
	MR-Egger	-0.027	-0.060 - 0.007	0.120	0.000	0.379

# Figure legend

Figure 1 Assumptions in Mendelian randomization analysis and study design.

Figure 1 Assumptions in Mendelian randomization analysis and study design.



Assumptions of Mendelian Randomization Analysis

- 1. Relevance assumption: Genetic instruments are associated with exposure
- 2. Independence assumption: Genetic instruments are not associated with any confounding that affect the exposure-outcome relationship
- 3. Exclusion restriction: Genetic instruments are assumed to affect the outcome only via the exposure